Author’s response to reviews

Title: Efficacy and safety of Acupuncture for chronic dizziness: a randomized, controlled clinical trial

Authors:

Zhe Xue (xuezhesanctity@163.com)
Cunzhi Liu (lcz623780@126.com)
Guangxia Shi (shiquiangxia2008@126.com)
Yan Liu (785344716@qq.com)
Zhao-Xin Li (lovevicky1111@163.com)
Zhen-Hua Zhang (zzh5988@qq.com)
Lin-Peng Wang (wlp5558@sina.com)

Version: 2 Date: 15 July 2013

Author’s response to reviews: see over
Dear editors,

Thank you for your time and efforts. According to the your comments on our manuscript, we have finished revisions.

Thank you again for your cooperation.

Sincerely yours,

Cun-Zhi Liu

Acupuncture and Moxibustion Department, Beijing Hospital of Traditional Chinese Medicine affiliated to Capital Medical University.